Author’s response to reviews

Title: Exploring the Parkinson patients' perspective on home-based video recording for movement analysis: a qualitative study

Authors:

Nienke de Vries (nienke.devries-farrouh@radboudumc.nl)
Katarzyna Smilowska (kasia.smilowska@gmail.com)
Joost Hummelink (joost_hummelink@hotmail.com)
Bronislaw Abramiuc (b.abramiuc@tue.nl)
Merel Van Gilst (m.m.v.gilst@tue.nl)
Bastiaan Bloem (bas.bloem@radboudumc.nl)
Peter De With (p.h.n.de.with@tue.nl)
Sebastiaan Overeem (s.overeem@tue.nl)

Version: 3 Date: 02 Oct 2018

Author’s response to reviews:

Dear Dr. Chen,

Thank you very much for reviewing our manuscript entitled “Exploring the Parkinson patients’ perspective on home-based video recording for movement analysis: a qualitative study”. We are happy to see that only reviewer 2 suggests some discretionary revisions, which we have provided in the present version of the manuscript.

The changes have been highlighted in one of the copies of the revised document. We hope that our revised manuscript can now be considered for publication in BMC Neurology.

We look forward to receiving your reaction!

With best wishes,

Nienke de Vries
Reviewer reports:

Walter Maetzler, MD (Reviewer 1): My comments have been adequately addressed.

We thank the reviewer for his positive appraisal of our revised manuscript.

Ming-Kuei Lu (Reviewer 2): There is a missing word ‘perceived motivation’ in Results (p. 5 line 43).

Some dots are missing (p.6 line 46; p.7 line 222, 28, 42; p.8 line 8).

‘the’ Netherlands instead of ‘The’ Netherlands (p. 9 line 33).

We thank the reviewer for noticing these issues. We have accommodated all suggestions in the new version of our manuscript.